Author’s response to reviews

Title: Corticosteroid dose increase is a risk factor for nonalcoholic fatty liver disease and contralateral osteonecrosis of the femoral head: A case report

Authors:

Hirokazu Shimizu (rottelfella.nnn@gmail.com)
Tomohiro Shimizu (simitom@wg8.so-net.ne.jp)
Daisuke Takahashi (rainbow-quest@pop02.odn.ne.jp)
Tsuyoshi Asano (tsuyosanny@yahoo.co.jp)
Ryuta Arai (ryutaaraisti@yahoo.co.jp)
Yasunari Takakuwa (yasunari.takakuwa@east.ntt.co.jp)
Norimasa Iwasaki (niwasaki@med.hokudai.ac.jp)

Version: 4 Date: 09 Feb 2019

Author’s response to reviews:

Response to Reviewers

We deeply appreciate the Editors for the valuable comments. We can try to give another quality to this paper. Responses to comments or suggestions are shown as following.

1. Please have the text edited by a professional language editing service or a native English speaking colleague. There are many issues with grammar, wording, spelling, and/or punctuation that need to be addressed before acceptance for publication.

Author’s response: Thank you for the important comment. We revised the text edited by the English language editing service (Editage).

2. Please combine your Discussion and Conclusions section under one heading, as per our editorial guidelines.

Author’s response: Thank you for pointing out. We revised the discussion section.
3. Thank you for providing your CARE checklist. However, at this stage it is not required and so we kindly ask that you remove it from your manuscript.

Author’s response: Thank you for pointing out. We removed it.

4. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Author’s response: Thank you for pointing out. We uploaded as the reviewer suggestion.

Sincerely,